Author's response to reviews

Title: Dextrocardia and asplenia in situs inversus totalis in an infant: a case report.

Authors:

ABNISH KUMAR Jr-III (abnish23@gmail.com)
MANOJ KUMAR SINGH MD (drmanuped@yahoo.com)
NEERAJ YADAV MD (neerajyadu@rediffmail.com)

Version: 3 Date: 4 July 2014

Author's response to reviews: see over
Author's response to reviews

Title: Dextrocardia and asplenia in situs inversus totalis in an infant: a case report.

Authors: abnish kumar (abnish23@gmail.com)

       manoj kumar singh (drmanuped@yahoo.com)

       neeraj yadav (neerajyadu@rediffmail.com)

Version: 2 Date: 05 july 2014

Author’s response to reviews: see over, changes made are indicated in blue.
A new paragraph was added to discussion section---

In our patient situs inversus was associated with asplenia and dextrocardia. Cardiac anomalies which was identified on echocardiography were ASD, VSD, AR, PS and TGA. Both kidneys were normal. In the vascular anomaly IVC and aorta both were on right side.
Reviewer's report

Title: Dextrocardia and asplenia in situs inversus totalis in an infant: a case report.

Version: 2 Date: 7 June 2014

Reviewer: Sameh Shehata

Which of the following best describes what type of case report this is?: New associations or variations in disease processes

Has the case been reported coherently?: Yes

Is the case report authentic?: Yes

Is the case report ethical?: Yes

Is there any missing information that you think must be added before publication?: No

Is this case worth reporting?: Yes

Is the case report persuasive?: Yes

Does the case report have explanatory value?: No

Does the case report have diagnostic value?: Yes

Will the case report make a difference to clinical practice?: Yes

Is the anonymity of the patient protected?: Yes

Comments to authors:

The photo quality and resolution of X-ray in Figure 1 needs to be improved.

High quality X-ray was uploaded.

Level of interest: An article of limited interest
Quality of written English: Acceptable

Declaration of competing interests:
None

Changed the 'Case Report' section header to 'Case Presentation'.

Header for the COMPETING INTEREST section was included.